Author's response to reviews

Title: Cognitive health among older adults in the United States and in England

Authors:

Kenneth M Langa (klanga@umich.edu)
David J Llewellyn (dl355@medschl.cam.ac.uk)
Iain A Lang (iain.lang@pms.ac.uk)
David R Weir (dweir@isr.umich.edu)
Robert B Wallace (robert-wallace@uiowa.edu)
Mohammed U Kabeto (mkabeto@umich.edu)
Felicia A Huppert (fah2@cam.ac.uk)

Version: 7 Date: 8 June 2009

Author’s response to reviews:

A problem with Table 4 was introduced in the last round of formatting changes. The header on the second part of the table needs to be included because the results for Models 1 through 4 need to be separated from the results for Models 5 through 8 (they were combined on the same page--p. 30--in the most recent version). I have re-inserted the header for Models 5 through 8, and separated them onto the next page. This is the only change made to this version of the manuscript. The manuscript is otherwise ready to publish from my standpoint.

As I noted in an e-mail sent with my last revision, my University would like to put out a press release timed with the on-line publication of this article. Can you please let me know whom we should contact to coordinate the publication date. Thank you for your help.